Author's response to reviews

Title: PRIMARY MELANOMA OF THE PROSTATE

Authors:

Tosev MD Georgi (georgi.tosev@med.uni-heidelberg.com)
Kuru MD Timur (tkuru@ukaachen.de)
Johannes PD Huber (Johannes.Huber@uniklinikum-dresden.de)
Gerald MD Freier (gerald.freier@med.uni-heidelberg.de)
Frank MD Bergmann (frank.bergmann@med.uni-heidekberg.de)
Jessica MD Hassel (jessica.hassel@med.uni-heidelberg.de)
Sascha PD Pahernik (sascha.pahernik@uni-heidelberg.de)
Markus Prof. Hohenfellner (marjus.hohenfellner@med.uni-heidelberg.de)
Boris Prof. Hadaschik (boris.hadaschik@med.uni-heidelberg.de)

Version: 6
Date: 24 March 2015

Author's response to reviews: see over
Dear editors,

we would like to thank you for accepting our manuscript „Primary melanoma of the prostate” for publication in BMC urology. Attached please find the changes you kindly asked for. Changes in the manuscript file are highlighted in yellow.

Kind regards from Heidelberg,

Georgi Tosev, MD
Department of Urology, University Hospital Heidelberg
Im Neuenheimer Feld 110, 69120 Heidelberg, Germany
Phone: +49 6221 56 39386; Fax: +49 6221 56 5366
Email: georgi.tosev@med.uni-heidelberg.de
EDITORIAL REQUEST:

1. ABSTRACT BACKGROUND NEEDS ADDITIONAL CONTEXT.
Thank you for the advice. We have rewritten the background accordingly.

2. INTRODUCTION NEEDS TO BE RE-NAMED BACKGROUND.
Thank you for the remark, correction was made.

3. ABBREVIATIONS, COMPETING INTERESTS, AUTHORS' CONTRIBUTIONS AND CONSENT STATEMENT ALL NEED TO BE MOVED TO BEFORE THE REFERENCES.
Thank you for the remark, corrections were made.

4. CLARIFY THAT CONSENT TO PUBLISH THE CASE REPORT WAS OBTAINED FROM THE NEXT-OF-KIN.
The initial consent was orally obtained from the patient and his family, unfortunately a written informed consent was not obtained from the patient for publication of this case report and any accompanying images since he died before the next contact. His next of kin agreed to the current version of the case report.

5. ACKNOWLEDGEMENT SECTION.
Thank you for the remark, corrections were made.